\begin{table}%T1
\caption{\label{table:sourcelist}Doppler boosting factors, Lorentz factor and viewing angles for all sources.} 
\small%\centering
\par
\begin{tabular}{lllcccrrrrrrrc}
 \hline
\hline\noalign{\smallskip}
B1950-name 	  & 	 Other name 	  & 	 Type 	  & 	 $\nu$	  & 	 $z$ 	  & 	 Quality 	  & 	 $\log(T_{\rm b})$ 	  & 	 $D_{\rm var}$ 	  & 	 $\beta_{\rm app}$ 	  & 	 $\Gamma_{\rm var}$ 	  & 	 $\theta_{\rm var}$ 	  & 	 $\log{R}$ 	  & 	 $P_{\rm max}$ 	  & 	 Ref. 	  \\
\noalign{\smallskip}\hline 
\object{0003$-$066} 	  & 	 NRAO 5 	  & 	 BLO 	  & 	 22 	  & 	 0.347 	  & 	 A 	  & 	 12.82 	  & 	 5.1 	  & 	 2.660 	  & 	 3.3 	  & 	 9.5 	  & --0.133 	  & 	 3.4 	  & 	 11 	  \\
 \object{0007+106} 	  & 	 III ZW 2 	  & 	 GAL 	  & 	 37 	  & 	 0.089 	  & 	 E 	  & 	 11.37 	  & 	 1.7 	  & 	 0.98 	  & 	 1.4 	  & 	 35.4 	  & 	 0.004 	  & 	 0.2 	  & 	 12 	  \\
 \object{0016+731} 	  & 	  	  & 	 LPQ 	  & 	 37 	  & 	 1.781 	  & 	 A 	  & 	 13.40 	  & 	 7.9 	  & 	 6.67 	  & 	 6.8 	  & 	 7.1 	  & --0.104 	  & 	 0.6 	  & 	 11 	  \\
 \object{0048$-$097} 	  & 	 PKS 0048-097 	  & 	 BLO 	  & 	 22 	  & 	 0.300 	  & 	 A 	  & 	 13.65 	  & 	 9.6 	  & 	 ... 	  & 	 ... 	  & 	 ... 	  & --0.082 	  & 	 27.1 	  & 	 2 	  \\
 \object{0059+581} 	  & 	  	  & 	 QSO 	  & 	 22 	  & 	 0.644 	  & 	 G 	  & 	 14.15 	  & 	 14.1 	  & 	 11.1 	  & 	 11.5 	  & 	 4.0 	  & 	 0.018 	  & 	 ... 	  & 	 ... 	  \\
 \object{0106+013} 	  & 	 OC 012 	  & 	 HPQ 	  & 	 22 	  & 	 2.099 	  & 	 E 	  & 	 14.50 	  & 	 18.4 	  & 	 26.16 	  & 	 27.8 	  & 	 2.9 	  & --0.172 	  & 	 7.1 	  & 	 11 	  \\
 \object{0133+476} 	  & 	 DA 55 	  & 	 HPQ 	  & 	 22 	  & 	 0.859 	  & 	 G 	  & 	 14.65 	  & 	 20.7 	  & 	 12.95 	  & 	 14.4 	  & 	 2.5 	  & --0.068 	  & 	 20.8 	  & 	 3 	  \\
 \object{0149+218} 	  & 	  	  & 	 LPQ 	  & 	 37 	  & 	 1.320 	  & 	 A 	  & 	 12.83 	  & 	 5.1 	  & 	 ... 	  & 	 ... 	  & 	 ... 	  & --0.024 	  & 	 ... 	  & 	 ... 	  \\
 \object{0202+149} 	  & 	 4C 15.05 	  & 	 HPQ 	  & 	 22 	  & 	 0.405 	  & 	 G 	  & 	 14.24 	  & 	 15.1 	  & 	 8.31 	  & 	 9.9 	  & 	 3.2 	  & --0.216 	  & 	 4.0 	  & 	 2 	  \\
 \object{0212+735} 	  & 	  	  & 	 HPQ 	  & 	 22 	  & 	 2.367 	  & 	 A 	  & 	 13.48 	  & 	 8.5 	  & 	 7.35 	  & 	 7.5 	  & 	 6.7 	  & --0.268 	  & 	 7.8 	  & 	 2 	  \\
 \object{0219+428} 	  & 	 3C 66A 	  & 	 BLO 	  & 	 37 	  & 	 0.444 	  & 	 A 	  & 	 11.92 	  & 	 2.6 	  & 	 ... 	  & 	 ... 	  & 	 ... 	  & 	 ... 	  & 	 29.7 	  & 	 5 	  \\
 \object{0224+671} 	  & 	  	  & 	 QSO 	  & 	 22 	  & 	 0.523 	  & 	 A 	  & 	 13.43 	  & 	 8.2 	  & 	 11.67 	  & 	 12.5 	  & 	 6.6 	  & --0.308 	  & 	 ... 	  & 	 ... 	  \\
 \object{0234+285} 	  & 	 4C 28.07 	  & 	 HPQ 	  & 	 37 	  & 	 1.213 	  & 	 G 	  & 	 14.32 	  & 	 16.1 	  & 	 12.23 	  & 	 12.7 	  & 	 3.4 	  & --0.090 	  & 	 11.3 	  & 	 2 	  \\
 \object{0235+164} 	  & 	  	  & 	 BLO 	  & 	 22 	  & 	 0.940 	  & 	 G 	  & 	 14.84 	  & 	 24.0 	  & 	 2.000 	  & 	 12.1 	  & 	 0.4 	  & --0.092 	  & 	 44.7 	  & 	 2 	  \\
 \object{0306+102} 	  & 	 PKS 0306+102 	  & 	 BLO 	  & 	 22 	  & 	 0.863 	  & 	 A 	  & 	 13.97 	  & 	 12.3 	  & 	 ... 	  & 	 ... 	  & 	 ... 	  & 	 ... 	  & 	 ... 	  & 	 ... 	  \\
 \object{0316+413} 	  & 	 3C 84 	  & 	 GAL 	  & 	 37 	  & 	 0.018 	  & 	 G 	  & 	 9.26 	  & 	 0.3 	  & 	 0.32 	  & 	 1.8 	  & 	 39.1 	  & --0.654 	  & 	 6.0 	  & 	 11 	  \\
 \object{0333+321} 	  & 	 NRAO 140 	  & 	 LPQ 	  & 	 22 	  & 	 1.263 	  & 	 G 	  & 	 14.74 	  & 	 22.2 	  & 	 12.7 	  & 	 14.7 	  & 	 2.2 	  & --0.140 	  & 	 0.7 	  & 	 12 	  \\
 \object{0336$-$019} 	  & 	 CTA 026 	  & 	 HPQ 	  & 	 22 	  & 	 0.852 	  & 	 A 	  & 	 14.42 	  & 	 17.4 	  & 	 22.32 	  & 	 23.0 	  & 	 3.2 	  & --0.133 	  & 	 19.4 	  & 	 2 	  \\
 \object{0355+508} 	  & 	 NRAO 150 	  & 	 QSO 	  & 	 37 	  & 	 1.510 	  & 	 G 	  & 	 14.05 	  & 	 13.1 	  & 	 ... 	  & 	 ... 	  & 	 ... 	  & --0.015 	  & 	 ... 	  & 	 ... 	  \\
 \object{0415+379} 	  & 	 3C 111 	  & 	 GAL 	  & 	 37 	  & 	 0.049 	  & 	 G 	  & 	 12.09 	  & 	 2.9 	  & 	 5.94 	  & 	 7.7 	  & 	 15.5 	  & --0.432 	  & 	 3.1 	  & 	 5 	  \\
 \object{0420$-$014} 	  & 	 OA 129 	  & 	 HPQ 	  & 	 37 	  & 	 0.915 	  & 	 E 	  & 	 14.59 	  & 	 19.9 	  & 	 7.52 	  & 	 11.4 	  & 	 1.9 	  & --0.097 	  & 	 26.2 	  & 	 11 	  \\
 \object{0430+052} 	  & 	 3C 120 	  & 	 GAL 	  & 	 22 	  & 	 0.033 	  & 	 G 	  & 	 13.01 	  & 	 5.9 	  & 	 5.42 	  & 	 5.5 	  & 	 9.7 	  & --0.488 	  & 	 0.5 	  & 	 5 	  \\
 \object{0440$-$003} 	  & 	 NRAO 190 	  & 	 HPQ 	  & 	 22 	  & 	 0.844 	  & 	 A 	  & 	 14.03 	  & 	 12.9 	  & 	 ... 	  & 	 ... 	  & 	 ... 	  & --0.291 	  & 	 12.5 	  & 	 11 	  \\
 \object{0458$-$020} 	  & 	 PKS 0458-020 	  & 	 HPQ 	  & 	 37 	  & 	 2.286 	  & 	 G 	  & 	 14.29 	  & 	 15.8 	  & 	 16.2 	  & 	 16.2 	  & 	 3.6 	  & --0.134 	  & 	 17.3 	  & 	 2 	  \\
 \object{0528+134} 	  & 	 PKS 0528+134 	  & 	 HPQ 	  & 	 22 	  & 	 2.070 	  & 	 E 	  & 	 15.18 	  & 	 31.2 	  & 	 18.73 	  & 	 21.2 	  & 	 1.6 	  & --0.059 	  & 	 4.1 	  & 	 5 	  \\
 \object{0552+398} 	  & 	 DA 193 	  & 	 LPQ 	  & 	 37 	  & 	 2.363 	  & 	 A 	  & 	 14.91 	  & 	 25.2 	  & 	 0.45 	  & 	 12.6 	  & 	 0.1 	  & --0.157 	  & 	 1.9 	  & 	 12 	  \\
 \object{0605$-$085} 	  & 	 PKS 0605-085 	  & 	 HPQ 	  & 	 37 	  & 	 0.872 	  & 	 A 	  & 	 13.34 	  & 	 7.6 	  & 	 19.98 	  & 	 30.2 	  & 	 5.0 	  & --0.191 	  & 	 9.8 	  & 	 11 	  \\
 \object{0642+449} 	  & 	 OH 471 	  & 	 LPQ 	  & 	 22 	  & 	 3.396 	  & 	 G 	  & 	 13.78 	  & 	 10.7 	  & 	 0.53 	  & 	 5.4 	  & 	 0.5 	  & --0.192 	  & 	 1.7 	  & 	 7 	  \\
 \object{0716+714} 	  & 	  	  & 	 BLO 	  & 	 22 	  & 	 0.310 	  & 	 E 	  & 	 13.81 	  & 	 10.9 	  & 	 10.22 	  & 	 10.3 	  & 	 5.2 	  & --0.219 	  & 	 17.3 	  & 	 4 	  \\
 \object{0735+178} 	  & 	 PKS 0735+17 	  & 	 BLO 	  & 	 37 	  & 	 0.424 	  & 	 G 	  & 	 12.43 	  & 	 3.8 	  & 	 ... 	  & 	 ... 	  & 	 ... 	  & --0.378 	  & 	 36.0 	  & 	 2 	  \\
 \object{0736+017} 	  & 	  	  & 	 HPQ 	  & 	 22 	  & 	 0.191 	  & 	 E 	  & 	 13.50 	  & 	 8.6 	  & 	 14.73 	  & 	 17.0 	  & 	 5.8 	  & --0.157 	  & 	 8.4 	  & 	 11 	  \\
 \object{0754+100} 	  & 	 OI 090.4 	  & 	 BLO 	  & 	 37 	  & 	 0.266 	  & 	 A 	  & 	 12.95 	  & 	 5.6 	  & 	 14.530 	  & 	 21.7 	  & 	 6.9 	  & --0.168 	  & 	 26.0 	  & 	 1 	  \\
 \object{0804+499} 	  & 	  	  & 	 HPQ 	  & 	 22 	  & 	 1.436 	  & 	 E 	  & 	 15.35 	  & 	 35.5 	  & 	 1.81 	  & 	 17.8 	  & 	 0.2 	  & --0.096 	  & 	 8.6 	  & 	 2 	  \\
 \object{0814+425} 	  & 	  	  & 	 BLO 	  & 	 22 	  & 	 0.245 	  & 	 G 	  & 	 12.69 	  & 	 4.6 	  & 	 1.720 	  & 	 2.7 	  & 	 8.4 	  & --0.139 	  & 	 8.7 	  & 	 2 	  \\
 \object{0827+243} 	  & 	 OJ 248 	  & 	 LPQ 	  & 	 37 	  & 	 0.941 	  & 	 A 	  & 	 14.05 	  & 	 13.1 	  & 	 21.3 	  & 	 23.9 	  & 	 3.9 	  & --0.037 	  & 	 1.5 	  & 	 7 	  \\
 \object{0836+710} 	  & 	 4C 71.07 	  & 	 LPQ 	  & 	 37 	  & 	 2.218 	  & 	 A 	  & 	 14.33 	  & 	 16.3 	  & 	 25.39 	  & 	 28.0 	  & 	 3.2 	  & --0.119 	  & 	 1.1 	  & 	 11 	  \\
 \object{0847$-$120} 	  & 	 J 0850-1213 	  & 	 QSO 	  & 	 37 	  & 	 0.566 	  & 	 G 	  & 	 14.35 	  & 	 16.5 	  & 	 ... 	  & 	 ... 	  & 	 ... 	  & 	 ... 	  & 	 ... 	  & 	 ... 	  \\
 \object{0851+202} 	  & 	 OJ 287 	  & 	 BLO 	  & 	 22 	  & 	 0.306 	  & 	 E 	  & 	 14.39 	  & 	 17.0 	  & 	 15.31 	  & 	 15.4 	  & 	 3.3 	  & --0.181 	  & 	 15.6 	  & 	 5 	  \\
 \object{0923+392} 	  & 	 4C 39.25 	  & 	 LPQ 	  & 	 37 	  & 	 0.695 	  & 	 G 	  & 	 12.60 	  & 	 4.3 	  & 	 42.94 	  & 	 216.1 	  & 	 2.6 	  & --1.757 	  & 	 0.5 	  & 	 2 	  \\
 \object{0945+408} 	  & 	 4C 40.24 	  & 	 HPQ 	  & 	 37 	  & 	 1.249 	  & 	 A 	  & 	 13.13 	  & 	 6.4 	  & 	 18.47 	  & 	 29.8 	  & 	 5.5 	  & --0.274 	  & 	 6.5 	  & 	 3 	  \\
 \object{0953+254} 	  & 	  	  & 	 LPQ 	  & 	 37 	  & 	 0.712 	  & 	 A 	  & 	 12.61 	  & 	 4.3 	  & 	 ... 	  & 	 ... 	  & 	 ... 	  & --0.268 	  & 	 2.2 	  & 	 2 	  \\
 \object{0954+658} 	  & 	 S4 0954+65 	  & 	 BLO 	  & 	 37 	  & 	 0.367 	  & 	 A 	  & 	 13.07 	  & 	 6.2 	  & 	 ... 	  & 	 ... 	  & 	 ... 	  & --0.054 	  & 	 33.7 	  & 	 11 	  \\
 \object{1055+018} 	  & 	 OL 093 	  & 	 HPQ 	  & 	 37 	  & 	 0.888 	  & 	 G 	  & 	 13.96 	  & 	 12.2 	  & 	 10.96 	  & 	 11.1 	  & 	 4.7 	  & --0.159 	  & 	 5.0 	  & 	 2 	  \\
 \object{1156+295} 	  & 	 4C 29.45 	  & 	 HPQ 	  & 	 22 	  & 	 0.729 	  & 	 E 	  & 	 15.06 	  & 	 28.5 	  & 	 24.85 	  & 	 25.1 	  & 	 2.0 	  & --0.080 	  & 	 9.2 	  & 	 12 	  \\
 \object{1219+285} 	  & 	 ON 231 	  & 	 BLO 	  & 	 22 	  & 	 0.102 	  & 	 A 	  & 	 10.96 	  & 	 1.2 	  & 	 ... 	  & 	 ... 	  & 	 ... 	  & --0.363 	  & 	 10.0 	  & 	 1 	  \\
 \object{1222+216} 	  & 	 PKS 1222+216 	  & 	 LPQ 	  & 	 37 	  & 	 0.432 	  & 	 A 	  & 	 12.85 	  & 	 5.2 	  & 	 21.12 	  & 	 45.5 	  & 	 5.1 	  & --0.210 	  & 	 1.5 	  & 	 9 	  \\
 \object{1226+023} 	  & 	 3C 273 	  & 	 LPQ 	  & 	 22 	  & 	 0.158 	  & 	 E 	  & 	 14.39 	  & 	 17.0 	  & 	 13.59 	  & 	 14.0 	  & 	 3.3 	  & --0.809 	  & 	 0.5 	  & 	 2 	  \\
 \object{1253$-$055} 	  & 	 3C 279 	  & 	 HPQ 	  & 	 37 	  & 	 0.536 	  & 	 E 	  & 	 14.84 	  & 	 24.0 	  & 	 20.64 	  & 	 20.9 	  & 	 2.4 	  & --0.242 	  & 	 39.2 	  & 	 5 	  \\
 \object{1308+326} 	  & 	 AU CV n 	  & 	 HPQ 	  & 	 22 	  & 	 0.997 	  & 	 G 	  & 	 14.26 	  & 	 15.4 	  & 	 27.500 	  & 	 32.2 	  & 	 3.2 	  & --0.096 	  & 	 25.0 	  & 	 2 	  \\
 \object{1324+224} 	  & 	  	  & 	 QSO 	  & 	 22 	  & 	 1.400 	  & 	 A 	  & 	 14.68 	  & 	 21.2 	  & 	 3.04 	  & 	 10.9 	  & 	 0.8 	  & --0.075 	  & 	 ... 	  & 	 ... 	  \\
 \object{1413+135} 	  & 	  	  & 	 BLO 	  & 	 22 	  & 	 0.247 	  & 	 G 	  & 	 13.96 	  & 	 12.2 	  & 	 1.860 	  & 	 6.3 	  & 	 1.4 	  & --0.112 	  & 	 13.0 	  & 	 8 	  \\
 \object{1418+546} 	  & 	 OQ 530 	  & 	 BLO 	  & 	 22 	  & 	 0.151 	  & 	 A 	  & 	 12.83 	  & 	 5.1 	  & 	 ... 	  & 	 ... 	  & 	 ... 	  & --0.175 	  & 	 19.0 	  & 	 1 	  \\
 \object{1502+106} 	  & 	 OR 103 	  & 	 HPQ 	  & 	 37 	  & 	 1.839 	  & 	 E 	  & 	 13.94 	  & 	 12.0 	  & 	 14.6 	  & 	 14.9 	  & 	 4.7 	  & --0.202 	  & 	 3.0 	  & 	 2 	  \\
 \object{1510$-$089} 	  & 	 PKS 1510-089 	  & 	 HPQ 	  & 	 37 	  & 	 0.360 	  & 	 E 	  & 	 14.37 	  & 	 16.7 	  & 	 20.29 	  & 	 20.7 	  & 	 3.4 	  & --0.167 	  & 	 7.8 	  & 	 11 	  \\
 \object{1538+149} 	  & 	 4C 14.60 	  & 	 BLO 	  & 	 37 	  & 	 0.605 	  & 	 A 	  & 	 12.60 	  & 	 4.3 	  & 	 8.750 	  & 	 11.2 	  & 	 10.5 	  & --0.296 	  & 	 17.4 	  & 	 2 	  \\
 \object{1606+106} 	  & 	 4C 10.45 	  & 	 LPQ 	  & 	 22 	  & 	 1.226 	  & 	 G 	  & 	 14.89 	  & 	 25.0 	  & 	 18.78 	  & 	 19.6 	  & 	 2.2 	  & --0.081 	  & 	 1.9 	  & 	 11 	  \\
 \object{1611+343} 	  & 	 DA 406 	  & 	 LPQ 	  & 	 37 	  & 	 1.401 	  & 	 A 	  & 	 14.11 	  & 	 13.7 	  & 	 14.17 	  & 	 14.2 	  & 	 4.2 	  & --0.179 	  & 	 2.3 	  & 	 6 	  \\
 \object{1633+382} 	  & 	 4C 38.41 	  & 	 HPQ 	  & 	 22 	  & 	 1.814 	  & 	 E 	  & 	 14.70 	  & 	 21.5 	  & 	 29.13 	  & 	 30.5 	  & 	 2.5 	  & --0.256 	  & 	 7.0 	  & 	 6 	  \\
 \object{1637+574} 	  & 	 OS 562 	  & 	 LPQ 	  & 	 22 	  & 	 0.751 	  & 	 A 	  & 	 14.13 	  & 	 14.0 	  & 	 10.61 	  & 	 11.0 	  & 	 4.0 	  & --0.093 	  & 	 1.5 	  & 	 12 	  \\
 \object{1641+399} 	  & 	 3C 345 	  & 	 HPQ 	  & 	 37 	  & 	 0.593 	  & 	 E 	  & 	 13.38 	  & 	 7.8 	  & 	 19.27 	  & 	 27.7 	  & 	 5.1 	  & --0.412 	  & 	 38.3 	  & 	 5 	  \\
 \object{1725+044} 	  & 	 PKS 1725+044 	  & 	 QSO 	  & 	 22 	  & 	 0.293 	  & 	 A 	  & 	 12.44 	  & 	 3.8 	  & 	 ... 	  & 	 ... 	  & 	 ... 	  & 	 ... 	  & 	 ... 	  & 	 ... 	  \\
 \object{1730$-$130} 	  & 	 NRAO 530 	  & 	 QSO 	  & 	 37 	  & 	 0.902 	  & 	 G 	  & 	 13.79 	  & 	 10.7 	  & 	 35.6 	  & 	 64.6 	  & 	 3.0 	  & --0.291 	  & 	 ... 	  & 	 ... 	  \\
 \object{1739+522} 	  & 	 S4 1739+52 	  & 	 HPQ 	  & 	 22 	  & 	 1.375 	  & 	 E 	  & 	 14.97 	  & 	 26.5 	  & 	 ... 	  & 	 ... 	  & 	 ... 	  & --0.040 	  & 	 3.7 	  & 	 11 	  \\
 \object{1741$-$038} 	  & 	 PKS 1741-038 	  & 	 HPQ 	  & 	 22 	  & 	 1.054 	  & 	 E 	  & 	 14.58 	  & 	 19.7 	  & 	 ... 	  & 	 ... 	  & 	 ... 	  & --0.055 	  & 	 9.2 	  & 	 2 	  \\
 \object{1749+096} 	  & 	 PKS 1749+096 	  & 	 BLO 	  & 	 37 	  & 	 0.322 	  & 	 E 	  & 	 13.93 	  & 	 12.0 	  & 	 5.880 	  & 	 7.5 	  & 	 3.8 	  & --0.030 	  & 	 31.3 	  & 	 2 	  \\
 \object{1803+784} 	  & 	 S5 1803+784 	  & 	 BLO 	  & 	 22 	  & 	 0.684 	  & 	 A 	  & 	 13.96 	  & 	 12.2 	  & 	 8.980 	  & 	 9.5 	  & 	 4.5 	  & --0.250 	  & 	 7.0 	  & 	 3 	  \\
  \hline
\object{1807+698} 	  & 	 3C 371.0 	  & 	 BLO 	  & 	 37 	  & 	 0.051 	  & 	 A 	  & 	 10.78 	  & 	 1.1 	  & 	 0.120 	  & 	 1.0 	  & 	 57.3 	  & --0.328 	  & 	 8.0 	  & 	 3 	  \\
 \object{1823+568} 	  & 	 4C 56.27 	  & 	 BLO 	  & 	 22 	  & 	 0.663 	  & 	 A 	  & 	 13.11 	  & 	 6.4 	  & 	 20.950 	  & 	 37.8 	  & 	 5.0 	  & --0.154 	  & 	 29.8 	  & 	 5 	  \\
 \object{1828+487} 	  & 	 3C 380 	  & 	 LPQ 	  & 	 37 	  & 	 0.692 	  & 	 A 	  & 	 12.97 	  & 	 5.7 	  & 	 13.66 	  & 	 19.3 	  & 	 7.1 	  & --0.391 	  & 	 0.4 	  & 	 2 	  \\
 \object{1928+738} 	  & 	 4C 73.18 	  & 	 LPQ 	  & 	 22 	  & 	 0.302 	  & 	 A 	  & 	 11.55 	  & 	 1.9 	  & 	 8.48 	  & 	 19.9 	  & 	 12.8 	  & --0.402 	  & 	 0.8 	  & 	 2 	  \\
 \object{1954+513} 	  & 	  	  & 	 LPQ 	  & 	 22 	  & 	 1.223 	  & 	 A 	  & 	 13.30 	  & 	 7.4 	  & 	 ... 	  & 	 ... 	  & 	 ... 	  & --0.181 	  & 	 1.4 	  & 	 11 	  \\
 \object{2005+403} 	  & 	  	  & 	 QSO 	  & 	 37 	  & 	 1.736 	  & 	 A 	  & 	 14.07 	  & 	 13.3 	  & 	 19.48 	  & 	 21.0 	  & 	 4.0 	  & --0.257 	  & 	 ... 	  & 	 ... 	  \\
 \object{2007+776} 	  & 	 S5 2007+77 	  & 	 BLO 	  & 	 22 	  & 	 0.342 	  & 	 A 	  & 	 13.39 	  & 	 7.9 	  & 	 ... 	  & 	 ... 	  & 	 ... 	  & --0.304 	  & 	 15.1 	  & 	 11 	  \\
 \object{2021+614} 	  & 	 OW 637 	  & 	 GAL 	  & 	 22 	  & 	 0.227 	  & 	 A 	  & 	 11.65 	  & 	 2.1 	  & 	 0.41 	  & 	 1.3 	  & 	 13.3 	  & 	 ... 	  & 	 0.3 	  & 	 3 	  \\
 \object{2022+171} 	  & 	  	  & 	 LPQ 	  & 	 22 	  & 	 1.050 	  & 	 A 	  & 	 12.99 	  & 	 5.8 	  & 	 ... 	  & 	 ... 	  & 	 ... 	  & 	 ... 	  & 	 ... 	  & 	 ... 	  \\
 \object{2121+053} 	  & 	  	  & 	 HPQ 	  & 	 22 	  & 	 1.941 	  & 	 A 	  & 	 14.25 	  & 	 15.3 	  & 	 13.05 	  & 	 13.2 	  & 	 3.7 	  & --0.061 	  & 	 10.7 	  & 	 7 	  \\
 \object{2134+004} 	  & 	 OX 057 	  & 	 LPQ 	  & 	 22 	  & 	 1.932 	  & 	 A 	  & 	 14.32 	  & 	 16.1 	  & 	 5.62 	  & 	 9.0 	  & 	 2.2 	  & --0.405 	  & 	 2.8 	  & 	 12 	  \\
 \object{2136+141} 	  & 	  	  & 	 LPQ 	  & 	 22 	  & 	 2.427 	  & 	 A 	  & 	 13.46 	  & 	 8.3 	  & 	 5.08 	  & 	 5.8 	  & 	 6.2 	  & --0.175 	  & 	 1.6 	  & 	 11 	  \\
 \object{2145+067} 	  & 	  	  & 	 LPQ 	  & 	 37 	  & 	 0.990 	  & 	 G 	  & 	 14.28 	  & 	 15.6 	  & 	 2.49 	  & 	 8.0 	  & 	 1.1 	  & --0.171 	  & 	 0.6 	  & 	 2 	  \\
 \object{2200+420} 	  & 	 BL LAC 	  & 	 BLO 	  & 	 37 	  & 	 0.069 	  & 	 E 	  & 	 13.28 	  & 	 7.3 	  & 	 10.700 	  & 	 11.6 	  & 	 7.3 	  & --0.264 	  & 	 23.0 	  & 	 1 	  \\
 \object{2201+315} 	  & 	 4C 31.63 	  & 	 LPQ 	  & 	 22 	  & 	 0.295 	  & 	 G 	  & 	 13.17 	  & 	 6.7 	  & 	 7.88 	  & 	 8.1 	  & 	 8.5 	  & --0.096 	  & 	 0.2 	  & 	 2 	  \\
 \object{2223$-$052} 	  & 	 3C 446 	  & 	 HPQ 	  & 	 22 	  & 	 1.404 	  & 	 G 	  & 	 14.31 	  & 	 16.0 	  & 	 16.470 	  & 	 16.5 	  & 	 3.6 	  & --0.203 	  & 	 17.3 	  & 	 10 	  \\
 \object{2227$-$088} 	  & 	  	  & 	 HPQ 	  & 	 22 	  & 	 1.562 	  & 	 A 	  & 	 14.31 	  & 	 15.9 	  & 	 4.95 	  & 	 8.8 	  & 	 2.0 	  & --0.137 	  & 	 6.9 	  & 	 12 	  \\
 \object{2230+114} 	  & 	 CTA 102 	  & 	 HPQ 	  & 	 37 	  & 	 1.037 	  & 	 E 	  & 	 14.28 	  & 	 15.6 	  & 	 15.51 	  & 	 15.5 	  & 	 3.7 	  & --0.269 	  & 	 10.9 	  & 	 7 	  \\
 \object{2234+282} 	  & 	  	  & 	 HPQ 	  & 	 22 	  & 	 0.795 	  & 	 A 	  & 	 13.03 	  & 	 6.0 	  & 	 ... 	  & 	 ... 	  & 	 ... 	  & --0.464 	  & 	 4.4 	  & 	 11 	  \\
 \object{2251+158} 	  & 	 3C 454.3 	  & 	 HPQ 	  & 	 37 	  & 	 0.859 	  & 	 E 	  & 	 15.26 	  & 	 33.2 	  & 	 14.86 	  & 	 19.9 	  & 	 1.3 	  & --0.564 	  & 	 16.0 	  & 	 1 	  \\
 \object{2254+074} 	  & 	 PKS 2254+074 	  & 	 BLO 	  & 	 22 	  & 	 0.190 	  & 	 A 	  & 	 11.91 	  & 	 2.5 	  & 	 ... 	  & 	 ... 	  & 	 ... 	  & 	 ... 	  & 	 21.0 	  & 	 1 	  \\
\hline
\end{tabular}
\smallskip
 References: (1) \cite{angel80}; (2) \cite{impey90}; (3) \cite{impey91}; (4) \cite{impey00}; (5) \cite{jorstad07}; (6)~\cite{lister00}; (7) \cite{moore84}; (8) \cite{punsly96}; (9) \cite{sluse05}; (10) \cite{smith85}; (11) \cite{stickel94}; (12) \cite{wills92}.
 \end{table}